Ocean and Coastal Management Shared visions for marine planning: Insights from Israel, South Africa and the United Kingdom --Manuscript Draft--

Manuscript Number:	OCMA-D-21-00848
Article Type:	Short Communication
Keywords:	Marine spatial planning; transdisciplinary; marine conservation; geospatial technologies; stakeholder engagement
Abstract:	The online symposium Shared Visions for Marine Spatial Planning: Insights from Israel, South Africa and the United Kingdom was held from 9-10 March 2021. Insights from this multi-disciplinary and international symposium included 1) current states of marine spatial planning (MSP) in the three countries, 2) how MSP can be a helpful tool to advance marine conservation, 3) the use and challenges of geospatial technologies for MSP, 4) how multidisciplinary, interdisciplinary and transdisciplinary efforts can help improve MSP processes and 5) recommendations for effective and collaborative MSP. Key reflections from the symposium included the need for MSP to be multi-, inter- and transdisciplinary in its stakeholder collaborations, aligned with in-country and area contexts.

Abstract:

The online symposium *Shared Visions for Marine Spatial Planning: Insights from Israel, South Africa and the United Kingdom* was held from 9-10 March 2021. Insights from this multi-disciplinary and international symposium included 1) current states of marine spatial planning (MSP) in the three countries, 2) how MSP can be a helpful tool to advance marine conservation, 3) the use and challenges of geospatial technologies for MSP, 4) how multidisciplinary, interdisciplinary and transdisciplinary efforts can help improve MSP processes and 5) recommendations for effective and collaborative MSP. Key reflections from the symposium included the need for MSP to be multi-, inter- and transdisciplinary in its stakeholder collaborations, aligned with in-country and area contexts.

Title: Shared visions for marine planning: Insights from Israel, South Africa and the United Kingdom

Key words: marine spatial planning; transdisciplinary; marine conservation; geospatial technologies; stakeholder engagement.

1. Introduction

Marine Spatial Planning¹ (MSP) has been adopted as a common process to achieve an integrated and ecosystem-based approach to manage the marine environment and uses therein, with rapid uptake globally (Kidd et al. 2020, Ehler 2020). With the inception of the UN Decade of Ocean Research and the pursuit of fulfilling the Sustainable Development Goals (SDGs), coastal states such as Israel, South Africa (SA) and the United Kingdom (UK) are now exploring best practices for the implementation of MSP². As a result, an online symposium entitled "Shared Visions for Marine Spatial Planning: Insights from Israel, South Africa and the United Kingdom" was held on 9-10 March, 2021. Insights from this symposium included 1) current states of marine spatial planning (MSP) in the three countries, 2) how multidisciplinary, interdisciplinary and transdisciplinary efforts can help improve MSP processes, 3) how MSP can advance marine conservation, 4) the use and challenges of geospatial technologies for MSP and 5) recommendations for effective and collaborative MSP. We provide a synthesis of the main outcomes of this symposium by including country-specific examples and recommendations (the Symposium is available online³).

2. Current states of MSP in Israel, SA and UK

In 2015 Technion - Israel Institute of Technology developed Israel's first marine spatial plan in response to the need to manage its Exclusive Economic Zone (EEZ) due to newly confirmed gas reserves (Portman 2015). Simultaneously, an initial government-led MSP process was initiated and published a plan in May, 2020 involving two development stages: 1) multidisciplinary analysis of the existing conditions and 2) defining policy principles for required regulation, planning and management of the maritime environment. This effort has already changed policy approaches for allocation of activities within the country's territorial waters, most significantly changes to the regulatory body that approves such plans - the Committee for the Protection of the Coastal Environment⁴ (Sas and Portman 2010).

MSP in South Africa had its inception in the 'National Environmental Management of the Oceans' (NEMO) white paper from 2014. Also in 2014, the Operation Phakisa "Unlocking the Ocean Economy" initiative was launched, aiming to unlock the economic potential of South Africa's oceans. As a result, a MSP act was fast tracked, while the NEMO white paper was not advanced. The MSP bill, finalized in 2017, outlines a framework that can enable a 'sustainable blue economy' whilst fostering socio-economic development (DEA 2018). In 2018 the MSP Act (MSP Act 2018) was gazetted and in April 2021 was signed into operation, providing mandatory requirements for the establishment of marine area plans (DEFF 2021). The Algoa Bay Project (ABP) in the Eastern Cape of South Africa is currently the first pilot site looking at the

³ https://portman.net.technion.ac.il/upcoming-conference-sustainable-governance-and-management-of-coasts-and-seas/

¹ Ehler and Douvere (2009) denote MSP as "a public process of analyzing and allocating the spatial and temporal distribution of human activities in marine areas to achieve ecological, economic, and social objectives that are usually specified through a political process". More current and critical definitions of MSP however acknowledge that MSP is, in fact, a political and social process informed by natural and social sciences where politics and power are inherent characteristics (Flannery et al. 2018; Ehler et al. 2019).

² Authors from Israel (Technion – Israel Institute of Technology) and the UK (University of Liverpool) were granted funding through the UK-Israel Inter-University Strategic Cooperation Programme (UIIUSCP) and then invited South Africa (Nelson Mandela University) to collaborate in order to engage with a global south country.

⁴ This committee has been in existence since 2004.

legislative, biophysical and socio-economic practicalities to inform the country's first MSP (Dorrington et al. 2018). It is a civil society-led initiative funded by the Government's Department of Science and Innovation through the National Research Foundation.

The legislative framework for MSP in the UK was formed by the 2009 Marine and Coastal Access Act, with more specific legislation for Scotland in the 2010 Marine (Scotland) Act. The 'UK Marine Policy Statement' sets out broad MSP terms and objectives throughout the UK which generally aligns with the European Union MSP process. MSP in the UK takes place independently in the four nations of England, Scotland, Wales and Northern Ireland. In England, marine plans are divided between inshore and offshore waters (covering internal and territorial waters and the EEZ respectively). England has 11 marine plan areas and as of June 2021, all outstanding marine plans were adopted⁵. Scotland has a two-tier system: a high level strategic national plan (2015) and 11 regional inshore plan areas at various stages of development. In Wales (2019) and Northern Ireland (draft 2018), a single plan covers the area for both inshore and offshore waters.

3. Multidisciplinary, interdisciplinary and transdisciplinary stakeholder collaborations in MSP

By design, MSP is a public process and requires multidisciplinary⁶, interdisciplinary⁷ and transdisciplinary⁸ approaches. These lead to better integration of efforts between use sectors as well as public buy-in by (Ansorg et al. 2020).

In Israel, MSP development has included consultation with various sectors such as shipping and trade, fisheries, gas exploration, heritage bodies and national parks. In order to integrate different interests and viewpoints into the MSP, a co-working steering committee was created to formulate policy and long-term strategies as well as coordinate processes with shared objectives. For example, through extensive research with different stakeholder groups on alternative protection scenarios and ecosystem service valuations, multi-sectoral perspectives can inform the zoning and planning of new Marine Protected Areas (MPAs) (Portman et al. 2016).

In South Africa, the lead authority for the development of MSP is the Department of Forestry, Fisheries and the Environment (DFFE)⁹ that engages with sectors from heritage, transport, mining, tourism and defense. Engagements with stakeholders will be led by the Marine Spatial Planning National Working Group (MSP NWG). However, government capacity to carry out equitable stakeholder processes needs to be leveled up for this process to be effective and just. Cooperation across sectors and disciplines have proven fruitful, such as in the ABP where Nelson Mandela University is investigating how best to facilitate cooperation between different disciplines and sectors towards the first multi-sectoral, ecosystem-based MSP in the country (Dorrington et al. 2018).

⁵ In the UK, a marine plan sets out how the MSP will be implemented in context-specific areas, or marine plan areas (MMO 2013).

⁶ "Multidisciplinarity draws on knowledge from different disciplines but stays within the boundaries of those fields" (Choi and Pak 2006).

⁷ "Interdisciplinarity analyzes, synthesizes and harmonizes links between disciplines into a coordinated and coherent whole" (Choi and Pak 2006).

⁸ "Transdisciplinarity claims to provide holistic schemes that subordinate disciplines and look at the dynamics of whole systems" (Alvargonza lez 2011). One of the defining features of transdisciplinarity is that it engages different sectoral stakeholders outside of academia.

⁹ This Department has undergone several name changes in recent years (from DEA to DEFF to DFFE)

In the UK, MSP is required by law¹⁰ to engage with a variety of stakeholders to understand their specific values. The marine planning authority in England, the Secretary of State for the Environment, is mandated to plan, implement, monitor and report through the Marine Management Organization (MMO). As a result, the MMO has integrated stakeholder engagement in every step of the MSP process.

4. Advancing marine conservation through MSP

MSP can advance marine conservation by prioritizing specific areas in need of biodiversity conservation, sustaining ecosystem services and identifying cumulative pressures on areas critical to socio-economic development or biophysical preservation (Foley et al. 2010). MSP can also offer different planning scenarios and assist in finding sustainable approaches to area-based ocean management (Portman 2015).

An Ecosystem Based Approach (EBA) is recommended as the most appropriate framework underpinning the development of MSP, where the health of marine environments are recognized as the foundation for preserving the system (Friedrich et al. 2020). To support EBA MSP, there is a need to envision and forecast the effects of management decisions on spatial and temporal outcomes under different scenarios, and to identify trade-offs between socio-economic and environmental goals (Foley et al. 2010). In South Africa, for example, system dynamics models are being developed to simulate temporal trends and sustainable outputs in selected marine uses, and to identify areas for management considerations in sectors to achieve a balance in social-ecological planning goals (Lombard et al. 2019b, Vermeulen et. al submitted).

MSP can also support other area-based management tools, such as MPAs, that may lack statutory power. In Israel for example, the co-development of a MPA management plan (2012) set a goal to protect 20% of Israel's territorial waters. However, since the plan held no statutory power, additional complementary processes were adopted through MSP, by allocating 9% territorial waters as no-take zones. Results from the process showed that no-take zones had substantial conservation benefits for vulnerable marine ecosystems as opposed to 'paper parks' which lack authority and regulation (Portman et al. 2016).

Ocean accounting is an important complementary tool in MSP to make strategic decisions affecting the ecological integrity of the ocean. The Blue Paper on National Accounting for the Ocean and Ocean Economy identifies that ocean accounting places an economic value on marine and coastal ecosystems and their services using metrics based on their impacts on "(1) real income and its distribution (and therefore social inclusivity), (2) ocean production (and economic metrics) and (3) changes in ocean wealth, including ecosystems" (Fenichel et al. 2020). Changes in ocean wealth are an important indicator of sustainability and can identify knowledge gaps for evidence-based ocean policy cycles and conservation plans linked to MSP.

5. Geospatial technologies as one of the tools to advance MSP

The oceans cover 71% of the Earth's surface, yet to date, only 20% of the seafloor has been measured by echo-sounders¹¹ (UNESCO 2020). Within the last two decades, a realization that the ocean environment represents a 'last frontier' and initiatives to better understand the structural layout of the earth's seafloor are being promoted at both an international (e.g., GEBCO Seabed 2030) and national (e.g. Operation

¹⁰ In the Marine and Coastal Access Act it states that 'interested persons' should be involved in the plan making process. Interested persons means - 'any person appearing to the marine plan authority to be likely to be interested in, or affected by, policies proposed to be included in a marine plan, and members of the general public' (MCAA 2009)

¹¹ Sonar used to determine the depth of water by transmitting acoustic waves.

Phakisa) level¹². Fourth Industrial Revolution advances in ocean monitoring and research technologies have greatly increased access to, resolutions and volumes of ocean data, in many cases within spatial realms (OECD 2016).

Technologies available to map the seafloor have vastly improved over recent years, and for MSP applications, sonar methods and geophysical mapping have been coupled with sampling or seafloor imaging campaigns that contribute to substrate maps. Specific geological and habitat boundaries are constructed using both supervised and unsupervised classification methods. Submersibles such as Autonomous Underwater Vehicles (AUVs) and Remotely Operated Vehicles (ROVs) are increasingly being applied as effective and efficient mapping tools (Sowers et al. 2020). It is anticipated that these maps can be used to model biological communities and produce benthic habitat maps for use in marine management. In South Africa scholars are both mapping the seafloor and developing algorithms that use machine learning to model benthic habitats

These geospatial technologies can also be used to map human activity to better understand the marine area and threats. The Blue Belt Programme in the UK, for example, has used a combination of satellite technologies and AIS to improve maritime domain awareness by assessing fleet distribution at different times of the year, creating 'heat maps' of shipping activity, bunkering and transshipments. This helps marine planners to determine where measures need to be put in place (e.g. Areas to be avoided).

Understanding the ocean and the natural processes occurring at the seafloor from a marine geological perspective is not without its limitations, primarily due to technological challenges in operating in this environment (Weatherall et al. 2015). For example, the technology used to map, observe, and understand land topography cannot penetrate more than tens of meters in ocean waters. Satellite measurements of ocean surface height provide a general view of the deep ocean floor through altimetry-derived predicted seafloor depths, but only to a limited extent (Cutter et al. 2003, McAdoo et al. 2004). Seafloor mapping remains an intensive and expensive task and has left most of our planet virtually unmapped.

6. Recommendations for effective and collaborative MSP

Following expert presentations and subsequent discussions, breakout sessions culminated in the formulation of several recommendations to be considered when working towards a shared vision of MSP across countries, sectors and disciplines. Support from the current literature is provided where relevant. We recommend:

- 1. Closing the science-to-policy gap through adaptive management: As ocean resource use, ecosystem services availability, social-ecological systems and scientific research programs are constantly changing, the development of dynamic and adaptive ocean management policies is required to respond to these changes (Winther et al. 2020, Maxwell et al. 2015, Portman 2016).
- 2. Interdisciplinary and transdisciplinary approaches: MSP is by definition interdisciplinary and should therefore be based on interdisciplinary collaborations and engage stakeholders and professionals through transdisciplinary collaborative processes from the beginning (Lombard et al 2019a, Grip and Blomqvist 2021).

¹² Without a level of certainty on habitat types it can be difficult to write prescriptive policies within a marine plan or assign areas to a particular activity if they are dependent on a specific habitat type.

- **3.** Data sharing policies, dialogue, transparency and deliberate collaboration among scientists and broader stakeholder groups need to be established to share data as a response to lack of data access and communication. This is also applied to technology use and development (Milanés Batista et al. 2020, Verutes et al. 2017).
- 4. Early and consistent stakeholder participation through empowered involvement in MSP development and ongoing commitment from convening authorities is required. This is vital to the success of MSP and involves clear communication aimed at creating a participatory management framework (Flannery et al. 2018, Bakker et al. 2019, Morf et al. 2019).
- **5.** Accommodation of different languages and education levels is required for MSP to be fair and equitable. MSP stakeholder engagement processes need to be accessible in all the languages of a given impacted or impacting community and for new concepts, goals and risks to be clearly communicated and discussed through objective knowledge brokers (Gorris 2019).
- 6. A diversity of incentive solution include state and non-state funding (economic); the establishment of collaborative platforms (participatory); penalties for non-compliance; improved stakeholder relations; increased political will; and effective legal enforcements (Ratsimbazafy et al. 2019, Bakker et al. 2019).
- 7. **Conflict resolution processes** should be components of MSP processes to help resolve sectoral conflict. Objective conflict resolution experts can guide deliberations to be constructive (Twichell et al. 2018, Morf et al. 2019).
- 8. **Political will** is required to build momentum and direct adequate resources and capacity towards effective MSP, especially during stakeholder engagement processes (Johnson, et al. 2020, Flannery et al. 2019).
- 9. **Top-down and bottom up approaches** are required to address both vertical and horizontal institutional and disciplinary barriers. For example, **strong leadership** is often required for successful MSP processes to help guide holistic long-term visions, whilst **participatory mapping**, when used correctly, can be a democratic tool for stakeholders to meaningfully inform MSP (Gaymer et al. 2014, Bakker et al. 2019).
- **10. Partnerships with the private sector** towards resource collaboration is beneficial (Johnson et al. 2020, Österblom et al. 2020).
- 11. Capacity building of management authorities and stakeholders, as well as social learning processes, need to be planned for in MSP processes (Gerhardinger et al. 2019, Ansong et al. 2019).

Figure 1 provides a visual artist's interpretation of what a shared MSP vision might encompass (drawn during the symposium plenary session).



Figure 1: Artist's interpretation of shared MSP visions (illustration by Efrat Goldberg).

7. Conclusion

Coastal nations can learn from one another regarding best practices for the development and implementation of MSP. However, to facilitate an adaptive, country-specific MSP process, the acknowledgement of contextual realities and integration of all local stakeholders is required. MSP provides a framework to manage the marine and coastal space, and also offers tools to advance marine conservation by applying ecosystem-based management, identifying priority areas for conservation, using accepted and novel ocean accounting frameworks and supporting management interventions that may lack statutory power.

The importance of transdisciplinarity and the early and consistent inclusion of all stakeholders impacted by and impacting on MSP is essential not only for the sustainability and adaptive ability of MSP but also to ensure a truly democratic process. An enabling environment characterized by political will, collaborative learning and investment in capacity building to enable stakeholders and implementers to engage equally and fairly were factors highlighted across all three country contexts. It is acknowledged that no country has all the enabling factors in place for the effective and just implementation of MSP but every effort should

be made to work towards these if MSP is to effectively manage how we use and conserve the ocean now and in the coming decades.

References

Alvargonzález, D. (2011). Multidisciplinarity, Interdisciplinarity, Transdisciplinarity, and the Sciences. International Studies in the Philosophy of Science, 25:4, 387-403. doi: 10.1080/02698595.2011.623366

Ansong, J., Calado, H. and Gilliland, P.M. (2019). A multifaceted approach to building capacity for marine/maritime spatial planning based on European experience. Marine Policy, 103422. doi:10.1016/j.marpol.2019.01.011

Ansong, J. O., McElduff, L., & Ritchie, H. (2020). Institutional integration in transboundary marine spatial planning: A theory-based evaluative framework for practice. *Ocean & Coastal Management*, 105430. doi: <u>https://doi.org/10.1016/j.ocecoaman.2020.105430</u>)

Bakker, Y. W., de Koning, J., and van Tatenhove, J. (2019). Resilience and social capital: The engagement of fisheries communities in marine spatial planning. Marine Policy, 99, 132-139. doi:10.1016/j.marpol.2018.09.032

Choi, B. C. K., and Pak, A. W. P. (2006). Multidisciplinarity, interdisciplinarity and transdisciplinarity in health research, services, education and policy: 1. Definitions, objectives, and evidence of effectiveness. Clinical and Investigative Medicine, 29:6, 351-364.

Cutter, G.R., Rzhanov, Y. and Mayer, L. A. (2003). Automated segmentation of seafloor bathymetry from multibeam echosounder data using local Fourier histogram texture features. Journal of Experimental Marine Biology and Ecology. 285, 355-370. doi: 10.1016/S0022-0981(02)00537-3

DEA (2018) *Marine Spatial Planning Bill*, Department of Environmental Affairs, Government Gazette No. 40726 of 28 March 2017

DEFF (2021). Commencement of the Marine Spatial Planning Act, 2018 (ACT No. 16 of 2018). Proclamation Notices. Government Gazette No. 44383, 1 April 2021.

Dorrington, R. A, Lombard, A. T., Bornman, T. G., Adams, J. B., Cawthra, H. C., Deyzel, S. H. B, Goschen, W. S., Liu, K., Mahler-Coetzee, J., Matcher, G. F., McQuaid, C., Parker-Nance, Paterson, A., Perissinotto, R.,Porri, F., Roberts, M., Snow, B. and Vrancken, P. (2018). Working together for our oceans: A marine spatial plan for Algoa Bay, South Africa. South African Journal of Science, 114:3/4, 1-6

Ehler, C. N. (2020). Two decades of progress in Marine Spatial Planning. Marine Policy, 104134. doi:10.1016/j.marpol.2020.104134

Ehler, C. and Douvere, F., (2009). Marine Spatial Planning: a step-by-step approach toward ecosystembased management. In: IOC Manual and Guides No. 53, ICAM Dossier No. 6. UNESCO,

Intergovernmental Oceanographic Commission and Man and the Biosphere Programme, (last accessed October, 27th 2018). http://unesdoc.unesco.org/images/0018/001865/186559e.pdf.

Ehler, C., Zaucha, J., Gee, K. (2019). Maritime spatial planning at the interface of research and practice. In: Zaucha, J., Gee, K. (Eds.), Maritime Spatial Planning, Past, Present, Future. London: Palgrave, 1-21

Fenichel, E. P., Milligan, B. and Porras, I. (2020). National Accounting for the Ocean and Ocean Economy, Washington, DC: World Resources Institute.

Flannery, W., Healy, N., & Luna, M. (2018). Exclusion and non-participation in Marine Spatial Planning. Marine Policy, 88, 32-40. doi:10.1016/j.marpol.2017.11.001

Flannery, W., Clarke, J. and McAteer, B. (2019). Politics and power in marine spatial planning. In Maritime Spatial Planning, 201-217. doi: 10.1007/978-3-319-98696-8_9

Foley, M. M., Halpern, B. S., Micheli, F., Armsby, M. H., Caldwell, M. R., Crain, C. M., Prahler, E., Rohr. N., Sivas, D., Beck, M. W., Carr, M. H., Crowder, L. B., Duffy, J. E., Hacker, S. D., McLeod, K. L., Palumbi, S. R., Peterson, C. H., Regan, H. M., Ruckelshaus, M. H., Sandifer, P. A. and Steneck, R. S. (2010). Guiding ecological principles for marine spatial planning. Marine Policy, 34:5, 955-966. doi:10.1016/j.marpol.2010.02.001

Friedrich, L. A., Glegg, G., Fletcher, S., Dodds, W., Philippe, M., and Bailly, D. (2020). Using ecosystem service assessments to support participatory marine spatial planning. Ocean & Coastal Management, 188. doi:10.1016/j.ocecoaman.2020.105121

Gaymer, C. F., Stadel, A. V., Ban, N. C., Cárcamo, P. F., Ierna, J., & Lieberknecht, L. M. (2014). Merging top-down and bottom-up approaches in marine protected areas planning: experiences from around the globe. Aquatic Conservation: Marine and Freshwater Ecosystems, 24:S2, 128-144. doi:10.1002/aqc.2508

Gerhardinger, L. C., Quesada-Silva, M., Gonçalves, L. R., & Turra, A. (2019). Unveiling the genesis of a marine spatial planning arena in Brazil. Ocean & Coastal Management, 179. doi:10.1016/j.ocecoaman.2019.104825

Gorris, P. (2019). Mind the gap between aspiration and practice in co-managing marine protected areas: A case study from Negros Occidental, Philippines. Marine Policy, 105, 12-19. doi:10.1016/j.marpol.2019.03.006

Grip, K. and Blomqvist, S. (2021). Marine spatial planning: coordinating divergent marine interests. Ambio. 50:6, 1172–1183. doi: 10.1007/s13280-020-01471-0

Johnson, A. E., McClintock, W. J., Burton, O., Burton, W., Estep, A., Mengerink, K., Tate, S. (2020). Marine spatial planning in Barbuda: A social, ecological, geographic, and legal case study. Marine Policy, 113. doi:10.1016/j.marpol.2019.103793 Kidd, S., Calado, H., Gee, K., Gilek, M. and Saunders, F. (2020). Marine Spatial Planning and sustainability: Examining the roles of integration - Scale, policies, stakeholders and knowledge. Ocean Coastal Management, 191:105182. doi: 10.1016/j.ocecoaman.2020.105182

Lombard, A. T., Ban, N. C., Smith, J. L., Lester, S. E., Sink, K. J., Wood, S. A., Jacob, A. L., Kyriazi, Z., Tingey, R. and Sims, H. E. (2019a). Practical Approaches and Advances in Spatial Tools to Achieve Multi-Objective Marine Spatial Planning. Frontiers in Marine Science, 6:166. doi: 10.3389/fmars.2019.00166

Lombard, A.T., Clifford-Holmes, J.K., Vermeulen, E., Witteveen, M. and Snow, B. (2019b). The advantages of system dynamics models in decision-support for integrated ocean management. In N. Pillay (Ed.), Proceedings of the Sixth Annual System Dynamics Conference in South Africa. 22-23 November 2018, Solomon Mahlangu House, University of the Witwatersrand, Johannesburg, South Africa. (pp. 94-103). South Africa System Dynamics Chapter. ISBN: 978-0-620-83145-1.

Maxwell, S. M., Hazen, E. L., Lewison, R. L., Dunn, D. C., Bailey, H., Bograd, S. J., Briscoe, D. K., Fossette, S., Hobday, A. J., Bennett, M., Benson, S., Caldwell, M. R., Costa, D. P., Dewar, H., Eguchi, T., Hazen, L., Kohin, S., Sippel, T. and Crowder, L. B. (2015). Dynamic ocean management: Defining and conceptualizing real-time management of the ocean. Marine Policy, 58, 42-50. doi: 10.1016/j.marpol.2015.03.014

McAdoo, B. G., Capone, M. K. and Minder, J. (2004). Seafloor geomorphology of convergent margins: implications for Cascadia seismic hazard. Tectonics, 23:6. doi: 10.1029/2003TC001570

Milanés Batista, C., Planas, J. A., Pelot, R., and Núñez, J. R. (2020). A new methodology incorporating public participation within Cuba's ICZM program. Ocean & Coastal Management, 186. doi:10.1016/j.ocecoaman.2020.105101

Marine Management Organisation (MMO) (August 2013). Strategic Scoping Report. Lancaster House, Hampshire Court, Newcastle upon Tyne, NE4 7YH, UK.

Morf, A., Moodie, J., Gee, K., Giacometti, A., Kull, M., Piwowarczyk, J., Schiele, K., Zaucha, J., Kellecioglu, I., Luttmann, A. and Strand, H. (2019). Towards sustainability of marine governance: Challenges and enablers for stakeholder integration in transboundary marine spatial planning in the Baltic Sea. Ocean & Coastal Management, 177, 200-212. doi:10.1016/j.ocecoaman.2019.04.009.

Marine and Coastal Access Act (MCAA) (2009) Marine plans: preparation and adoption. Statement of Public Participation. (c.23) Schedule 6(5).

MSP Act. (2018). Marine Spatial Planning Act No. 16 of 2018. Government Gazette No. 42444, 6 May 2018.

OECD (2016), The Ocean Economy in 2030, OECD Publishing, Paris. http://dx.doi.org/10.1787/9789264251724-en

Österblom, H., Cvitanovic, C., van Putten, I., Addison, P., Blasiak, R., Jouffray, J-B., Bebbington, J., Hall, J., Ison, S., LeBris, A., Mynott, S., Reid, D. and Sugimoto, A. (2020). Science-industry collaboration: sideways or highways to ocean sustainability? One Earth, 3:1, 79-88. doi: 10.1016/j.oneear.2020.06.011.

Portman, M. E. (2015). Marine spatial planning in the Middle East: Crossing the policy-planning divide. Marine Policy, 61, 8-15. doi:10.1016/j.marpol.2015.06.025

Portman, M. E. (2016). *Environmental Planning for Oceans and Coasts: Methods, Tools, Technologies*. Switzerland: Springer.

Portman, M. E., Shabtay-Yanai, A. and Zanzuri, A. (2016). Incorporation of socio-economic features' ranking in multicriteria analysis based on ecosystem services for marine protected area planning. PLoS ONE, 11:5, 1-17. doi: 10.1371/journal.pone.0154473

Ratsimbazafy, H., Lavitra, T., Kochzius, M., & Hugé, J. (2019). Emergence and diversity of marine protected areas in Madagascar. Marine Policy, 105, 91-108. doi:10.1016/j.marpol.2019.03.008

Sas, E., Fishhendler, I., & Portman, M. E. (2010). The demarcation of arbitrary boundaries for coastal zone management: the Israeli case. *Journal of Environmental Management*, *91*, 2358-2369.

Sowers, D., Masetti, G., Mayer, L. A., Johnson, P., Gardner, J. V., and Armstrong, A. A., 2020. Standardized Geomorphic Classification of Seafloor Within the United States Atlantic Canyons and Continental Margin, Frontiers in Marine Science 7(9), 1-9.

Twichell, J., Pollnac, R., & Christie, P. (2018). Lessons from Philippines MPA Management: Social Ecological Interactions, Participation, and MPA Performance. Environment Management, 61:6, 916-927. doi:10.1007/s00267-018-1020-y

UNESCO, (2020): https://en.unesco.org/news/nearly-fifth-worlds-ocean-floor-now-mapped

Vermeulen, E.A., J.K. Clifford-Holmes, U. Scharler, A.T. Lombard. An Exploratory System Dynamics Model to Support Marine Spatial Planning in Algoa Bay, South Africa. Submitted to Environmental Modelling and Software June 2021.

Verutes, G. M., Arkema, K. K., Clarke-Samuels, C., Wood, S. A., Rosenthal, A., Rosado, S., Canto, M., Bood, N. and Ruckelshaus, M. (2017). Integrated planning that safeguards ecosystems and balances multiple objectives in coastal Belize. International Journal of Biodiversity Science, Ecosystem Services & Management, 13:3, 1-17. doi:10.1080/21513732.2017.1345979

Weatherall, P., Marks, K.M., Jakobsson, M.; Schmitt, T., Tani, S., Arndt, J.E., Rovere, M., Chayes, D., Ferrini, V., Wigley, R. (2015). A new digital bathymetric model of the world's oceans. Earth Space Science, 2, 331–345. doi:10.1002/2015EA000107

Winther, J., Dai, M., Rist, T., Hoel, A., Li, Y., Trice, A., Morrissey, K., Juinio-Manez, M., Fernandes, L., Unger, S., Scarano, F., Halpin, P. and Whitehouse, S. (2020). Integrated ocean management for a sustainable ocean economy. Nature Ecology & Evolution, 4, 1451-1458. doi:10.1038/s41559-020-1259-6